Author's response to reviews

Title: Delayed diagnosis of spontaneous extrusion of a renal calculus in an adult male with spina bifida and paraplegia: case report. Measures to be taken to reduce medical errors in spinal cord injury patients

Authors:

Subramanian Vaidyanathan (vaidyanathansiuhotmail.com)
Peter L Hughes (Peter.Hughes@mail.soh-tr.nwest.nhs.uk)
Bhakul M Soni (Bhakul.Soni@mail.soh-tr.nwest.nhs.uk)
Gurpreet Singh (gurpreet.singh@ukgateway.net)
Paul Mansour (Paul.Mansour@mail.soh-tr.nwest.nhs.uk)
Pradipkumar Sett (Pradip.Sett@mail.soh-tr.nwest.nhs.uk)

Version: 2 Date: 26 Nov 2001

Dear Editor
Figure 2-A and Figure 2-B appear to create confusion. Therefore, I numbered the figures 1, 2, 3, 4, 5, and 6.
Thanks
S Vaidyanathan